Author's response to reviews

Title: An indeterminate mucin-producing cystic neoplasm containing an undifferentiated carcinoma with osteoclast-like giant cells: a case report of a rare association of pancreatic tumors.

Authors:

Marco Chiarelli (mee.chiarelli@gmail.com)
Angelo Guttadauro (angelo.guttadauro@unimib.it)
Martino Gerosa (martinogerosa@libero.it)
Alessandro Marando (alessandromarando@gmail.com)
Francesco Gabrielli (francesco.gabrielli4@tin.it)
Matilde De Simone (matilde.desimone@alice.it)
Ugo Cioffi (ugocioffi5@gmail.com)

Version: 3 Date: 21 August 2015

Author's response to reviews:

A point by point to reviewers comment

1) The English Language has been revised as suggested
2) Last sentence in the conclusion section has been modified
3) The legends have been corrected as suggested

Sincerely Ugo Cioffi